Author's response to reviews

Title: Analyses of associations between three positionally cloned asthma candidate genes and asthma or asthma-related phenotypes in a Chinese population

Authors:

Huanyu Zhou (huanyuzhou@hrca.harvard.edu)
Xiumei Hong (xiumei.hong@gmail.com)
Shanqun Jiang (shanqun@gmail.com)
Hongxing Dong (rstar126900@gmail.com)
Xiping Xu (xipingxu18@gmail.com)
Xin Xu (xin_xu@harvard.edu)

Version: 6 Date: 21 November 2009

Author's response to reviews:

Dear Editor,

We have checked over our manuscript carefully and made some small typographical changes. Now the manuscript is acceptable for publication.

Please let me know if you have any questions.

Xin Xu, MD, PhD
Mark and Catherine Winkler Assistant Professor
Harvard School of Public Health
665 Huntington Avenue, FXB-101
Boston, MA 02115
Tel. (617)4324572
Fax. (617)4322956
Email: xin_xu@harvard.edu